Author’s response to reviews

Title: Prevalence of low back pain and occupational risk factors among the Chinese coal miners

Authors:

Guangxing Xu (xuguangxing2005122@163.com)
Liping Li (lpli@stu.edu.cn)
Fengying Liu (huiyiyun123@sina.com)
Desheng Pei (bailiazhu@126.com)
Sheng Wang (shengw@bjmu.edu.cn)
Dong Pang (dong.pang@hotmail.com)

Version: 3 Date: 26 August 2011

Author’s response to reviews: see over
Dear Editors:

On behalf of my co-authors, I am submitting the enclosed material “Prevalence of low back pain and occupational risk factors among the Chinese coal miners” for possible publication in your journal. The abstract have been published in your journal. All authors have read and approved this version of the article, and due care has been taken to ensure the integrity of the work. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal.

We believe the paper may be of particular interest to the readers of your journal as it the study firstly reported the prevalence of low back pain in Chinese coal miners and the role of occupational risk factors.

Description of the changes made, please view:

1. Remove the entire first page of the manuscript.
2. Separate the sections of the abstract in the submission system.
3. Add the context information in the background section of both abstracts.
4. In the methods section, research carried out on humans must be in compliance with the Helsinki Declaration. The study was not experimental research.

Correspondence should be addressed to Liping Li at the following address, phone and email address:

Prof. Liping Li

Injury Prevention Research Centre

Medical College of Shantou University
22 Xinling Road, Shantou, Guangdong 515041, PRC
Tel: 0086-75488900467
Email: lpli@stu.edu.cn
Thank you very much for your attention to our paper.
Sincerely yours,
Liping Li